Author's response to reviews

Title: Prevalence of HIV and syphilis infections among pregnant women attending antenatal clinics in Tanzania

Authors:

Joel Manyahi (manyahijoel@yahoo.com)
Boniphace Jullu (jullub@gmail.com)
Mathias Abuya (mattyabuya@yahoo.com)
Bonita Kilama (bonitakilama@yahoo.com)
Veryeh Sambu (veryeh77@hotmail.com)
James M Juma (jji162us@gmail.com)
Joseph Nondi (josenondi@yahoo.co.uk)
Bernard Rabiel (benrabiels@yahoo.co.uk)
Mecky I Matee (mateemeky@yahoo.com)
Geoffrey Somi (geoffreysomi@gmail.com)
Joel Ngayongeje (ndayongejej@yahoo.co.uk)

Version: 8 Date: 23 April 2015

Author's response to reviews: see over
Dear Editor

BMC Public Health

Please find enclosed manuscript entitled “Prevalence of HIV and syphilis infections among pregnant women attending antenatal clinics in Tanzania, 2011” which we are resubmitting after addressing all editors’ comments.

Authors would like to appreciate all constructive comments which have been provided and we have addressed all comments.

We hope the revised manuscript is now acceptable for publication in BMC Public Health.

On behalf of all authors,

Yours sincerely,

Joel Manyahi

Response to editor

Editor's comments:

Please format your "Author's contribution" section. Please mention each of the author's contribution for the manuscript. Please use initials when referring to the authors. For example, Joel Manyahi, JM provided the concept of the study.

Author's response: We appreciate your comments and we have managed to address all the comments.